Author's response to reviews

Title: Sustained remission of rheumatoid arthritis with SSRI antidepressant: a case report and a brief review of literature

Authors:

Rajeev Krishnadas (rk60s@clinmed.gla.ac.uk)
Ranjit Krishnadas (ranjit.krishnadas@gmail.com)
Jonathan Cavanagh (jonathan.cavanagh@climed.gla.ac.uk)

Version: 3 Date: 21 October 2010

Author's response to reviews: see over
Dear Prof Kidd,

Sub: Revision of manuscript “Sustained remission of rheumatoid arthritis with SSRI antidepressant: a case report” for consideration of publication

Thank you for accepting this article for publication in its current form. Please find the revised version of the manuscript. Changes made have been recommended by the independent editorial production team.

1. I have made an addition to the title, which now reads – “Sustained remission of rheumatoid arthritis with SSRI antidepressant: a case report and a brief review of literature”.

2. Patient ID has been stated consistently. The change has been made in the main text.

3. References have been modified to contain up to 30 authors. However, I must note that I had formatted the initial draft using Endnote, which uses “et al” even to numbers less than 30.

4. Minor revisions have been made as suggested. However, I could not find a section numbering.

Thanking you

Rajeev Krishnadas
Clinical Lecturer in Psychological Medicine

Sackler Institute of Psychobiological Research

Section of Psychological Medicine

Southern General Hospital

Glasgow

G51 4TF